Author’s response to reviews

Title: Genetic variants in FBLIM1 gene do not contribute to SAPHO syndrome and chronic recurrent multifocal osteomyelitis in typical patient groups

Authors:
Gunter Aßmann (gunter.assmann@uks.eu)
Michaela Köhm (Michaela.Koehm@ime.fraunhofer.de)
Volker Schuster (Volker.Schuster@medizin.uni-leipzig.de)
Frank Behrens (Frank.Behrens@ime.fraunhofer.de)
Rotraut Mössner (rmoessn@gwdg.de)
Nina Magnolo (Nina.Magnolo@ukmuenster.de)
Vinzenz Oji (Vinzenz.Oji@ukmuenster.de)
Harald Burkhardt (Harald.Burkhardt@kgu.de)
Ulrike Hüffmeier (Ulrike.Hueffmeier@uk-erlangen.de)

Version: 5 Date: 25 Apr 2020

Author’s response to reviews:

Dear Editor,

we submit a new version of the manuscript that contains links and more information in the methods section and availability of data section on the database of the Genome Aggregation Database (gnomAD). We used it to assess frequency differences between our cases and the general population.

Please let us know whether the manuscript is now acceptable for publication.

Looking forward to hearing from you!

Yours sincerely

Ulrike Hüffmeier on behalf of all authors